**Author's response to reviews**

**Title:** Predictors of the development of myocarditis and acute renal failure in patients with leptospirosis: An observational study

**Authors:**

Dinesh L B Dassanayake (dlbdassanayake@gmail.com)
Harith Wimalaratna (harithlc@slt.net.lk)
Damith Nandadewa (d.nandadewa@gmail.com)
Asanka Nugaliyadda (asankanuga@yahoo.com)
Champa N Ratnatunga (Champa26@gmail.com)
Suneth B Agampodi (sunethagampodi@yahoo.com)

**Version:** 4  **Date:** 13 January 2012

**Author's response to reviews:** see over
Dr D.L.B.Dassanayake,
Registrar in medicine,
Teaching Hospital,
Kandy,
Sri Lanka.
13th January 2012

The Editor,
BMC Infectious Disease
Dear Sir

Submitting the proof read manuscript “Predictors of the development of myocarditis or acute renal failure in patients with leptospirosis: An observational study “for publication in BMC Infectious Disease

Many thanks for reviewing our manuscript. There were two typographical errors which were corrected. Here are the corrections made in point form.

1. In results section of the abstract data were in two formats, I have converted them to number format in the second sentence of the section.
2. The name of the hospital was missing in the Methods section (study setting) of the main manuscript (page 3), which was added in the first sentence of the section.

Looking forward to see the article in published form.

Thank you.
Dr D L B Dassanayake,
(Corresponding author)
Registrar in medicine,
dlbdassanayake@gmail.com